

# Mindfulness-based cognitive therapy (MBCT) reduces the association between depressive symptoms and suicidal cognitions in patients with a history of suicidal depression.

Barnhofer, Thorsten; Crane, Catherine; Brennan, Kate; Duggan, Danielle; Crane, Rebecca S.; Eames, Catrin; Radford, Sholto; Silverton, Sarah; Fennell, Melanie; Williams, J. Mark G.

# Journal of Consulting and Clinical Psychology

DOI: 10.1037/ccp0000027

Published: 12/09/2015

Peer reviewed version

Cyswllt i'r cyhoeddiad / Link to publication

Dyfyniad o'r fersiwn a gyhoeddwyd / Citation for published version (APA): Barnhofer, T., Crane, C., Brennan, K., Duggan, D., Crane, R. S., Eames, C., Radford, S., Silverton, S., Fennell, M., & Williams, J. M. G. (2015). Mindfulness-based cognitive therapy (MBCT) reduces the association between depressive symptoms and suicidal cognitions in patients with a history of suicidal depression. *Journal of Consulting and Clinical Psychology*, 83(6), 1013-1020. https://doi.org/10.1037/ccp0000027

#### Hawliau Cyffredinol / General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

• Users may download and print one copy of any publication from the public portal for the purpose of private

study or research.
You may not further distribute the material or use it for any profit-making activity or commercial gain
You may freely distribute the URL identifying the publication in the public portal ?

#### Take down policy

This article may not exactly replicate the final version published in the APA journal. It is not the copy of record.

#### Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Mindfulness-Based Cognitive Therapy (MBCT) Reduces the Association between Depressive Symptoms and Suicidal Cognitions in Patients with a History of Suicidal Depression

Thorsten Barnhofer, Catherine Crane, Kate Brennan, Danielle Duggan University of Oxford, UK

Rebecca S. Crane, Catrin Eames, Sholto Radford, Sarah Silverton Bangor University, UK

> Melanie J. V. Fennell, & J. Mark G. Williams University of Oxford, UK

#### Author Note

Thorsten Barnhofer, Catherine Crane, Kate Brennan, Danielle Duggan, Melanie J. V. Fennell, J. Mark G. Williams, Department of Psychiatry, Warneford Hospital, Oxford, OX3 7JX; Rebecca S. Crane, Catrin Eames, Sholto Radford, Sarah Silverton, Centre for Mindfulness Research and Practice, Bangor University, Dean St Building, Bangor, LL57 2PZ, UK.

Thorsten Barnhofer is now at Freie Universitat Berlin, Habelschwerdter Allee 45, 14195 Berlin; Germany. Kate Brennan is now at Queen's University, University Road, Belfast, United Kingdom, BT7 1NN. Correspondence concerning this article should be addressed to Thorsten Barnhofer, Freie Universitaet Berlin, Habelschwerdter Allee 45, 14195 Berlin, Germany. Email: thorsten.barnhofer@fu-berlin.de.

## **Acknowledgments and Author Declarations**

Analyses of the current study were based on data from the Staying Well after Depression trial. We gratefully acknowledge the contribution of the people and institutions involved in conducting this trial: the administrative team in Oxford and Bangor, and the staff of the North Wales Organisation for Randomised Trials in Health (NWORTH) for their assistance with administration and data management, the general practice and mental health professionals who referred patients to the study, the National Institute for Health Research Mental Health Research Network, the Clinical Research Centre of the National Institute for Social Care and Health Research, the Thames Valley Primary Care Research Network, Gwynedd & Mon Local Health Boards and Gwynedd County Council (North Wales), who supported fieldwork. We thank our participants for giving time to take part in the trial. The trial was funded by Wellcome Trust Grant GR067797, awarded to J. Mark G. Williams and Ian T, Russell (Trial Registration Number: ISRCTN97185214). The funders had no role in study design, in the collection, analysis and interpretation of data, in the writing of this report, and in the decision to submit the article. All authors declare financial support for the submitted work from the Wellcome Trust. J. Mark G. Williams, Melanie J. V. Fennell, Rebecca S. Crane, Sarah Silverton, and Thorsten Barnhofer, and/or their institutions have received honoraria or fees for lectures, workshops, courses, and/or educational presentations on mindfulness or mindfulnessbased cognitive therapy. J. Mark G. Williams, Rebecca S. Crane and Sarah Silverton have received royalties for books on MBCT.

#### Abstract

*Objective:* In patients with a history of suicidal depression, recurrence of depressive symptoms can easily reactivate suicidal thinking. This study investigated whether training in mindfulness, which is aimed at helping patients "decenter" from negative thinking, may help to weaken the link between depressive symptoms and suicidal cognitions.

*Method:* Analyses were based on data from a recent randomized controlled trial, in which previously suicidal patients were allocated to MBCT, an active control treatment, Cognitive Psycho-Education (CPE), which did not include any meditation practice, or Treatment-as-Usual (TAU). We compared the associations between depressive symptoms, as assessed through self-reports on the Beck Depression Inventory-II, and suicidal thinking, as assessed through the Suicidal Cognitions Scale, after the end of the treatment phase.

*Results:* In patients with minimal to moderate symptoms at the time of assessment, comparisons of the correlations between depressive symptoms and suicidal cognitions showed significant differences between the groups. While suicidal cognitions were significantly related to levels of symptoms in the two control groups, there was no such relation in the MBCT group.

*Conclusions:* The findings suggest that, in patients with a history of suicidal depression, training in mindfulness can help to weaken the association between depressive symptoms and suicidal thinking, and thus reduce an important vulnerability for relapse to suicidal depression.

Keywords: Suicidality, depression, mindfulness, recurrence, cognitive reactivity

*Public Health Significance Statement:* This study suggests that Mindfulness-Based Cognitive Therapy can help to uncouple the link between suicidal thinking and depressive symptoms in patients who have suffered from suicidal depression in the past. This is important, as a strong coupling between depressive symptoms and suicidal cognitions has been suggested to increase risk for relapse to suicidal depression. It remains unclear from the current study, however, how stable these changes are and in how far they translate into actual reductions in rates of relapse. Mindfulness-Based Cognitive Therapy (MBCT) Reduces the Association between Depressive Symptoms and Suicidal Cognitions in Patients with a History of Suicidal Depression

Suicidality often recurs with depressive episodes (Ten Have et al., 2009). In fact, of all the non-core symptoms of depression (symptoms other than negative mood and anhedonia), suicidality has been found to be the one that reemerges most consistently: if patients have been suicidal during a given episode of depression, they are very likely to become suicidal again the next time they are depressed (Williams, Crane, Barnhofer, Van der Does, & Segal, 2006). Even if not resulting in suicidal behavior, suicidality represents a serious outcome in its own right. Recurrence of suicidal ideation has worrying wider implications, as risk of detrimental outcomes increases with increasing number of previous episodes of suicidality (Van Orden et al., 2010). It is of paramount importance, therefore, to find interventions that can prevent escalation of depressive symptoms into suicidality. MBCT (Segal, Williams, & Teasdale, 2002) has been shown to be effective in reducing risk for relapse and recurrence in depression (Piet & Hougaard, 2011). Here we were interested more specifically to investigate whether the treatment can help to reduce vulnerability for relapse to suicidal depression.

In individuals who have been depressed in the past, maladaptive patterns of thinking can easily become reactivated through only minor triggers such as subtle changes in mood (Segal et al., 2006; Segal, Gemar, & Williams, 1999; Scher, Ingram, & Segal, 2005). It has been suggested that the propensity for this cognitive reactivity develops as a consequence of associative learning during previous episodes of depression (Segal, Williams, Teasdale, & Gemar, 1996). As depressed patients repeatedly engage in negative thinking, associations between negative mood and the patterns of thinking that are prevalent in this mood are formed and strengthened. The particular content and pattern that is being associated with negative mood is dependent on what is part of the "rehearsal pool". Following initial suggestions by Beck (1996), Rudd (2000, 2006) has introduced the concept of a suicidal mode, referring to a cognitive-affective-behavioral network that, once activated, guides

synchronous processes that manifest on a number of different levels including suicidal cognitions, negative affect, physiological arousal, and motivation to engage in suicidal behaviour. In line with this view, research has shown that, under conditions of low mood, previously depressed patients with a history of suicidality are significantly more likely to show suicidal thinking and related cognitive deficits than previously depressed patients without such a history (Williams, Barnhofer, Crane, & Beck, 2005; Williams, Van der Does, Barnhofer, Crane, & Segal, 2008). According to Rudd (2000), the cognitive component of the suicidal mode is characterized by three categories of core beliefs: unlovability, helplessness, and poor distress tolerance. Activation of these beliefs, and ensuing perceptions of helplessness, inadequacy, and inability to cope are assumed to increase risk for suicidal ideation and behavior. Recent research has demonstrated that cognitions related to these three categories of core beliefs significantly predict suicidal ideation over and above depressive symptoms and general levels of hopelessness (Ellis & Rufino, 2014; Rudd et al., in press).

MBCT has been specifically designed to help participants become better able to recognize and disengage from maladaptive patterns of thinking. Training in mindfulness aims to serve this purpose by cultivating a metacognitive mode that allows patients to 'decenter' from negative thinking, i.e. to observe their "thoughts and feelings as temporary, objective events in the mind, as opposed to reflections of the self that are necessarily true" (Fresco et al., 2007). Consistent with this rationale, research into the mechanisms of MBCT for the prevention of relapse to depression has demonstrated that increases in the ability to decenter significantly mediate effects of the treatment on rates of relapse (Bieling et al., 2012). The aim of the current analyses was to investigate whether mindfulness training can provide an effective tool for weakening the link between depressive symptoms and suicidal cognitions related to the core beliefs of the suicidal mode described by Rudd (2000).

In order to test this hypothesis, we analyzed data from a randomized controlled trial of MBCT for relapse prevention that included a large number of participants with a history of

suicidal depression. Participants in this trial had to have a history of at least three previous episodes of Major Depression. As risk of relapse is known to increase with number of previous episodes (Mueller et al., 1999; Solomon et al., 2000), use of this criterion ensured that participants were likely to be at high risk of relapse. It has been suggested that cognitive vulnerability processes become increasingly easier to trigger with repeated recurrence of depression (Segal et al., 1996), and that in order to prevent relapse in individuals with a history of repeated episodes, mental training is of particular importance. Previous trials of MBCT have found that the training has significant preventative effects in patients with three or more previous episodes, while there is little evidence for beneficial effects in patients with two or less previous episodes (Piet & Hougaard, 2011). The Staying Well after Depression Trial (Williams et al., 2010) compared the effects of MBCT to two control conditions: an active control treatment, Cognitive Psycho-Education (CPE), in which participants received all of the elements of MBCT except the meditation practice, and treatment as usual (TAU). Patients reported levels of suicidal cognitions related to the core beliefs of unlovability, helplessness and poor distress tolerance, measured using the Suicidal Cognitions Scale by Rudd, Joiner and Rajab (2001), assessed before and after the treatment phase. This provided the opportunity to investigate effects on the association between current depressive symptoms and suicidal cognitions. We hypothesized that following MBCT, suicidal reactivity, as indicated by correlations between symptoms of depression and suicidal cognitions, would be significantly lower than following treatment with CPE (which included verbal discussion of decentering but no mindfulness *practice*) and TAU, which was unlikely to include any systematic training of or reference to decentering strategies.

#### Method

The study protocol for the trial was approved by the local ethics committee (Oxfordshire Research Ethics Committee C; July 2008). All participants provided written consent prior to any research activity.

#### **Participants and study flow**

We assessed eligibility using the Structured Clinical Interview for DSM-IV, Research Version (SCID; First, Spitzer, Gibbon, & Williams, 2002), which was conducted by formally trained research assistants. Inclusion criteria for the trial were (a) age between 18 and 70 years, (b) a history of at least three previous episodes of depression, meeting DSM-IV TR criteria, two of which must have occurred within the last 5 years and one within the last 2 years, and (c) being in remission during the previous 8 weeks. Potential trial participants were deemed *not* to be in recovery or remission, and hence *ineligible*, if they reported that at least one week during the previous 8 they had experienced *either* a core symptom of depression (depressed mood, anhedonia) *or* suicidal feelings plus at least one other symptom of depression. For the current analysis, focusing on recurrence of suicidal thinking, we included only those participants who had reported a previous history of suicidal ideation or suicidal behaviour as assessed by SCID interview.

Exclusion criteria were (a) history of schizophrenia, schizoaffective disorder, bipolar disorder, current abuse of alcohol or other substances, organic mental disorder, pervasive developmental disorder, or regular non-suicidal self-injury, (b) positive continuing response to cognitive behavior therapy (CBT), due to the known effects of CBT in reducing risk of relapse (Hollon, Stewart, & Strunk, 2006), which otherwise could have potentially confounded treatment effects in the trial; (c) current psychotherapy or counseling more than once a month, (d) regular meditation practice (meditating more than once per month); or (e) inability to complete research assessments through difficulty with English, visual impairment, or cognitive difficulties. As part of the trial, interviewer reliability for SCID diagnoses of

7

depression was assessed by independent ratings of a sample of 91 follow-up interviews conducted by two independent psychiatrists, which yielded an agreement of  $\kappa = 0.74$ , 95% CI [0.60, 0.87] between the original assessor and the independent rater.

Participants provided information on all measures described in this study at baseline (pre-treatment), following which they were randomly assigned to one of the three treatment arms: MBCT, CPE or TAU. Patients in MBCT and CPE received an individual pre-class interview and eight weekly 2-hour sessions; patients in TAU received an initial interview session, in which the importance of treatment seeking was stressed. Most baseline assessments were scheduled within a three-month window between treatment cohorts prior to the beginning of the interventions (M = 58.79, SD = 31.38 days), most post-treatment assessments within a 6-week window following the end of the interventions (M = 29.52, SD = 19.53). MBCT followed the standard protocol (Segal et al., 2002) except for greater emphasis on patterns and thoughts associated with suicidality and its escalation, and preparation of an action plan to respond to suicidal crisis. CPE comprised all elements of the MBCT program except the experiential cultivation of mindfulness through meditation practice. Details of the trial protocol are given in Williams et al. (2010); the main results of the two-centre trial are reported in (Williams et al., 2014.

#### **Outcome measures**

Severity of depressive symptoms was measured using the Beck Depression Inventory-II (BDI-II). Current levels of suicidal cognitions were measured using the Suicidal Cognitions Scale (SCS).

Beck Depression Inventory-II (BDI-II; Beck, Steer, & Brown, 1996). The BDI-II consists of 21 groups of statements, referring to the presence of symptoms of depression over the past 2 weeks. Internal consistency in the current sample was  $\alpha = .88$  at pre-treatment, and .90 at post-treatment.

Suicidal Cognitions Scale (SCS; Rudd, et al., 2001). The SCS consists of questions referring to core beliefs reflecting the suicidal mode of thinking described by Rudd (2000). Factor analyses conducted by Rudd et al. (in press) have indicated factors related to the suicidal schemas of unbearability (e.g., "I can't stand this pain anymore") and unlovability (e.g. "I am completely unworthy of love"). Recent studies by other authors (Bryan et al., 2014; Ellis & Rufino, 2014) suggest a third factor reflecting unsolvability ("Suicide is the only way to end this pain"), which is in line with the initial theoretical suggestions by Rudd (2000). Items of the self-report questionnaire are rated on a Likert-type scale scored from 1 to 5. In the current study we used the 20-item version of the scale that was published by Rudd et al. (2001). This version differs from a recently revised version of the questionnaire, in which Rudd et al. (in press) removed items containing the word "suicide" (items 2 and 15, which both load highly on the unsolvability factor identified by Ellis and Rufino, 2014, and Bryan et al., 2014) in order to reduce construct overlap for studies predicting suicidal ideation and behavior. Results in our study remained unchanged when analyses were based on this revised version. All analyses focused on the total scale. Internal consistency of this scale in the current sample was  $\alpha = .93$  at pre-treatment, and .94 at post-treatment, which is comparable to findings from previous studies (Ellis & Rufino, 2014; Rudd et al., in press). Ellis and Rufino (2014) reported a test-retest reliability of r = .46, p < .01, over the course of treatment in a sample of psychiatric in-patients. In the current sample, test-retest reliability from pre- to post-treatment was r(192) = .50, p < .001.

## Results

### **Demographic and clinical characteristics**

Information on patient demographics and clinical characteristics is presented in Table 1. Of the 274 individuals who participated in the trial, 221 reported past suicidality (ideation or behavior), 194 of whom had provided full data sets at both pre- and post-assessment. Seventy-seven of these participants had been assigned to MBCT, 78 to CPE and 39 to TAU.

There were no significant differences between the three treatment groups in terms of sociodemographic variables or course characteristics (see Table 1, all p > .05). At entry into the study, patients in this trial sub-sample had a mean age of M = 43.7 years (SD = 12.1). Seventy-four % were female, and 95% were of Caucasian ethnicity. Clinical course of the disorder was characterized by a mean age of onset of M = 20.1 (SD = 10.1). Participants had suffered from an average of M = 7.3 (SD = 5.5) previous episodes of depression. Those who had dropped out of treatment showed no significant differences in sociodemographic variables and course characteristics from those who had provided data at post-treatment (all p> .10).

# **Scatterplots and Frequency Distributions**

Inspection of the scatterplots of the relation between BDI-II and SCS scores at posttreatment (see Figure 1, upper panel) showed a number of cases with extreme values that would have disproportionately determined linear relations. We decided to deal with these outliers by excluding participants with severe levels of symptoms, using the conventional upper boundary for moderate symptoms on the BDI-II, a score of 29, as a cut-off. This is in keeping with the theoretical focus of the study on suicidal reactivity, or more specifically on the question of whether treatment can reduce the association between depressive symptoms and suicidal cognitions that becomes obvious already with presence of minimal to moderate symptoms of depression. There were 4 participants in the MBCT group, 5 in the CPE group, and 2 in TAU with scores above 29, indicating comparable rates of outliers across the groups, 6 %, 6 %, and 5 %, respectively. All of these participants showed high levels of suicidal cognitions, suggesting that, in these cases, the treatments had neither prevented relapse to depression nor to suicidality. Participants with severe levels of symptoms at post-treatment showed significantly higher numbers of previous episodes, *F*(1, 165) = 4.85, *p* = .02,  $\eta^2$  = .02 (*M* = 11.22, *SD* = 11.30, in participants with BDI-II scores above 29, *M* = 7.10, *SD* = 4.97, in participants with BDI-II scores of 29 or lower), but did not differ with regard to all other sociodemographic and course characteristics (all p < .10).

Inspection of the scatterplots and frequency distributions also showed that there was a considerable number of participants in each of the groups who showed no or extremely low levels of symptoms. As activation of suicidal cognitions is unlikely to occur when patients show no or only extremely low levels of symptoms, inclusion of these participants could have artificially inflated linear relations between symptoms and suicidal cognitions. For this reason, we decided to run analyses including only participants with a BDI-II score of 3 or higher, indicating presence of at least one symptom at full expression over the past two weeks. In order to explore robustness of findings, we then re-ran analyses including participants with scores of at least 2, 1, and, eventually, 0. In the MBCT group, there were 21 participants (27%) with a BDI-II score of 2 or less, 17 participants (22%) with a score of 1 or less, and 10 (13%) with a score of 0. In the CPE group, there were 19 participants (24%) with a score of 2 or less, 17 participants (21%) with a score of 1 or less, and 12 (15%) with a score of 0. In the TAU group, 13 participants (33%) had a score of 2 or less, 11 (28%) had a score of 1 or less, and 5 (12%) had a score of 0. Participants with a score of less than 3 on the BDI-II did not differ significantly in sociodemographic and course characteristics from those with BDI-II scores of 3 or more (all p < .10, apart from a trend for age of onset, F(1, 192) = 3.12, p = .07,  $\eta^2$  = .01; participants with BDI-II less than 3: M = 17.96 (SD = 7.86), participants with BDI-II of 3 or more: M = 20.87 (SD = 10.78)). The lower panel of Figure 1 depicts the scatterplot of the relation between BDI-II and SCS scores for the more restricted sample of participants who reported minimal to moderate symptoms of depression, i.e. BDI-II scores  $\geq 3$ and  $\leq 29$ .

# **Changes in BDI and SCS scores**

Means and standard deviations of BDI-II and SCS scores at pre- and post-assessment in participants with minimal to moderate symptoms are listed in Table 2. A 2 (time) × 3 (group)

repeated measures ANOVA of BDI-II scores showed no significant effects for group, time or their interaction (all p > .10). An identical analysis for SCS scores showed a significant main effect for group, F(1, 129) = 16.88, p = .00,  $\eta^2 = .11$ , which was qualified by a significant group by time interaction, F(2, 129) = 4.2, p = .02,  $n^2 = .06$ , due to significant reductions in the MBCT group,  $M_{(I-J)} = 8.53$ , SE = 1.61, p = .00, while there were no significant differences in the CPE group,  $M_{(I-J)} = 2.93$ , SE = 1.58, p = .07, and the TAU group,  $M_{(I-J)} = 1.74$ , SE =2.28, p = .44. These results remained substantially unchanged when analyses were conducted with changes in BDI-II scores entered as a covariate, thus suggesting that changes in SCS scores were not entirely due to reductions in depressive symptoms. When the above analyses were conducted using BDI-II lower cut-offs of 2 and 1, respectively, follow-up analyses also showed significant reductions in SCS scores in the CPE group. When analyses were conducted including participants with a score of 0, the time by treatment interaction for SCS scores reached only trend levels, F(2, 179) = 2.83, p = .06,  $\eta^2 = .03$ , due to relatively more pronounced reductions in the two control groups,  $M_{(I-J)} = 2.98$ , SE = 1.76, p = .09, in the TAU group, and,  $M_{(I-J)} = 3.15$ , SE = 1.26, p = .01, in the CPE group, compared to  $M_{(I-J)} = 6.95$ , SE =1.25, p = .00, in the MBCT group.

Changes in SCS scores in the MBCT group showed a marginally significant correlation with the amount of meditation practice during the time of the intervention (mean time of daily formal meditation practice reported by the participants), r(50)= -.26, p = .05.

### Investigation of group differences in the relation between BDI-II and SCS scores

In order to investigate the relation between post-treatment BDI-II and SCS scores within the three groups, we computed zero-order correlations. There were significant associations in the CPE group, r(52) = .60, p = .00, and in the TAU group, r(24) = .45, p = .02, but no significant association in the MBCT group, r(50) = .06, p = .67. Fisher *r*-to-*z* transformations indicated significant differences between correlations in the MBCT and the CPE group, z = 3.19, p = .000, and the MBCT and the TAU group, z = 1.67, p = .04, while

there was no significant difference between correlations in the CPE and TAU groups, z = .86, p = .19.

When we computed correlations in groups defined by lower boundaries for BDI-II inclusion scores, we found slightly higher correlations in the MBCT group r(54) = .11 for BDI-II  $\ge 2$ , r(61) = .18 for BDI-II  $\ge 1$ , and r(71) = .30 for BDI  $\ge 0$ , while correlations in the CPE group, r(54) = .60 for BDI-II  $\ge 2$ , r(59) = .61 for BDI-II  $\ge 1$ , and r(71) = .65 for BDI-II  $\ge 0$ , and the TAU group, r(35) = .50 for BDI-II  $\ge 2$ , r(30) = .50 for BDI-II  $\ge 1$ , and r(24) =.52 for BDI-II  $\ge 0$ , remained at the same level. The pattern of differences remained largely unchanged, except that *r*-to-*z* transformations reflecting differences between correlations in the MBCT group and the TAU group reached only trend levels in groups with a lower boundary of 1, z = -1.62, p = .05, or 0, z = -1.32, p = .09.

The fact that the difference between correlations in the MBCT group and the TAU group was less robust may, at least partly, have been due to the smaller size of the TAU group resulting from the recruitment ratio of 2:2:1. As we did not assume significant differences in the relation between BDI-II and SCS scores in the CPE and in the TAU group, we re-ran analyses comparing the MBCT group with both of the control groups together. This yielded a correlation coefficient of r(78) = .56, p = .00, in the control groups, which was significantly different from the coefficient of r(50) = .06, p = .67, found in the MBCT group, z = -3.13, p = .00. The difference between correlations in the two groups remained significant when tested across all the different lower boundaries for BDI-II inclusion scores, MBCT: r(54) = .11, control groups: r(86) = .56, z = -2.98, p = .00, for BDI  $\ge 2$ ; MBCT: r(61) = .18, control groups: r(91) = .57, z = -2.78, p = .00, for BDI  $\ge 1$ ; MBCT: r(71) = .30, control groups: r(108) = .61, z = -2.60, p = .00, for BDI  $\ge 0$ .

#### Discussion

It has been suggested that, in patients with a history of suicidal depression, changes in mood can reactivate a suicidal mode of processing, reflected in cognitions revolving around core beliefs of unlovability, helplessness, and poor distress tolerance. In order to investigate whether training in mindfulness can uncouple this link, the current study investigated the relations between levels of depressive symptoms and levels of suicidal cognitions in different treatment groups of a randomized-controlled trial. In line with our hypothesis, there was a significantly weaker association between levels of symptoms and suicidal cognitions in the group who had received MBCT than in the groups who had received treatments that did not include training in mindfulness. Furthermore, despite the fact that levels of depressive symptoms remained relatively unchanged participants in the MBCT group showed a general reduction in suicidal cognitions, suggesting that in participants of this group suicidal cognitions.

Interpretation of these findings needs to take into account a number of limitations. Importantly, the study assessed the association of depressive symptoms and suicidal cognitions at a single predetermined point in time, which may or may not have represented the most relevant period with regard to risk for recurrence. Indeed variation at this point was such that we had to exclude a considerable number of participants who showed no or only very low levels of symptoms and would thus have been very unlikely to experience activation of a suicidal mode. Because the distribution of levels of symptoms in remitted samples tends to be skewed with lower numbers of individuals showing relatively higher levels of depression, relatively high sample sizes are needed in order to demonstrate robust relations, which is likely to account for our observation that the difference between the MBCT and the CPE group was more robust than the difference between the MBCT and the TAU group (recruited to be only half the size of the other two groups).

Furthermore, analyses excluded participants who showed severe levels of symptoms. While there was no reason to assume that cases with severe levels of symptoms did not represent true observations, inspection of the scatterplots within the MBCT group identified these cases as outliers that would have disproportionately determined linear relations. Exclusion of these cases is in line with the focus of the study on vulnerability to relapse and suicidal reactivity as compared to the question of whether relapse to depression continues to be associated with suicidality. As the current analyses focused on effects at the end of treatment, it remains unclear to which degree the relapse of these participants was due to adverse effects or ineffectiveness of the treatment, or the fact that presence of severe symptoms might have precluded effective participation in the training.

It is also important to note that the current study was based on a sample of participants who had been in full remission at entry into the trial. Many patients with a recurrent course of the disorder show residual symptoms between episodes (Judd, Akiskal, & Maser, 1998; Paykel et al., 1995) and it is possible that this criterion might have excluded those who are most reactive. This is likely to have worked against our chances of finding strong effects. In fact, previous research has found that MBCT produced better effects in patients with unstable remissions as compared to those with stable remissions (Segal et al., 2010).

Because of the use of a dismantling design, in which MBCT was compared to a control treatment, CPE, that included all the elements of MBCT except the mindfulness training it is possible to attribute the observed differences in association between MBCT and CPE as specifically due to the effects of mindfulness practice. Reactivation of suicidal cognitions can occur relatively automatically, and this finding supports the assumption that systematic mental training is needed in order to help patients to become better at decentering from the habitual maladaptive patterns of thinking that arise as a consequence of cognitive reactivity.

Increasing the ability to decenter is one of the main aims of MBCT and the current findings suggest that participants can learn to effectively transfer this skill to respond to suicidal cognitions. This is important, as past research on cognitive interventions has shown that positive effects in alleviating depression do not necessarily transfer to reduce vulnerability for suicidality (Cuijpers et al., 2013). Interestingly, the version of MBCT used in the current trial did not include any major modifications to specifically address suicidality. Instead, the training followed the general principle of working with the material that occurs spontaneously during meditations assuming that spontaneous occurrence of difficulties provides the most important indicator of their relevance. Because of this, the extent to which participants had opportunities for learning and practicing new responses to suicidality may have differed widely between participants and therapy groups. It is possible, therefore, that inclusion of exercises that address more specifically and explicitly the problem of suicidal reactivity might further increase the potential of MBCT to reduce vulnerability for suicidal depression. While decreases in coupling between depressive symptoms and suicidal cognitions should reduce risk of relapse to suicidal depression, it was not possible to test relations between reactivity and relapse in the current data as variation in symptoms between participants precluded inferences about individual differences in reactivity. Comparisons between groups showed no significant differences in rates of relapse to suicidal depression at the end of the follow-up of the trial in patients who had reported a history of suicidality (MBCT: n = 12 (13%), CPE: n = 18 (20%), TAU: n = 11 (25%),  $\chi^2 (2, N = 221) = 2.78$ , p - .24).

With regard to the scope of the current findings, it is important to keep in mind that the current study investigated reactivation of suicidal cognitions in response to minimal to moderate levels of depression, and in a group of participants with a history of suicidality that, in most cases, was characterized by suicidal ideation, but not behavior and attempts. While mindfulness may also be helpful in the context of more significant levels of symptoms and in patients with more severe histories of suicidality, such contexts often require significantly different approaches to those used in the current context. For example, dialectical behavior therapy, which was developed to treat people with borderline personality disorder and chronic suicidality (Linehan, Armstrong, Suarez, Allmon, & Heard, 1991), offers mindfulness training as an adjunct to a broad range of other interventions aimed at dealing with emotional instability, destructive impulsivity, and chaotic life problems.

Altogether, the current results suggest that, within the context of relapse prevention, mindfulness training can effectively uncouple the link between depressive symptoms and suicidal cognitions in individuals who are at risk of suicidality, and thus reduce an important vulnerability for relapse to suicidal depression. Further research will have to investigate how stable the described effect is, and in how far it translates into actual reductions in rates of relapse.

#### References

- Beck, A. T. (1996). Beyond belief: a theory of modes, personality, and psychopathology. In P.M. Salkovskis (Ed.), *Frontiers of cognitive therapy* (pp. 1 25). New York, NY:Guilford Press.
- Beck, A. T., Steer, R. A., & Brown, G. K. (1996). *Manual for the Beck Depression Inventory-II*. San Antonio, TX: Psychological Corporation.
- Bieling, P. J., Hawley, L. L., Bloch, R. T., Corcoran, K. M., Levitan, R. D., Young, L. T., ...
  Segal, Z. V. (2012). Treatment-specific changes in decentering following mindfulnessbased cognitive therapy versus antidepressant medication or placebo for prevention of depressive relapse. *Journal of Consulting and Clinical Psychology*, 80(3), 365–72. doi:10.1037/a0027483
- Bryan, C. J., Rudd, M. D., Wertenberger, E., Etienne, N., Ray-Sannerud, B. N., Morrow, C.
  E., ... Young-McCaughon, S. (2014). Improving the detection and prediction of suicidal behavior among military personnel by measuring suicidal beliefs: An evaluation of the Suicide Cognitions Scale. *Journal of Affective Disorders*, 159, 15–22.
- Cuijpers, P., de Beurs, D. P., van Spijker, B. A. J., Berking, M., Andersson, G., & Kerkhof,
  A. J. F. M. (2013). The effects of psychotherapy for adult depression on suicidality and
  hopelessness: a systematic review and meta-analysis. *Journal of Affective Disorders*,
  144(3), 183–90. doi:10.1016/j.jad.2012.06.025
- Ellis, T. E., & Rufino, K. A. (2014). A psychometric study of the Suicide Cognitions Scale in psychiatric patients. *Assessment*, Advance online publication. http://dx.doi.org/10.1037/pas0000028
- First, M. D., Spitzer, R. L., Gibbon, M., & Williams, J. B. W. (2002). Structured Clinical Interview for DSM-IV TR Axis I Disorders, Research Version. New York, NY: Biometrics Research, New York State Biometric Institute.

- Fresco, D. M., Moore, M. T., van Dulmen, M. H. M., Segal, Z. V, Ma, S. H., Teasdale, J. D., & Williams, J. M. G. (2007). Initial psychometric properties of the experiences questionnaire: validation of a self-report measure of decentering. *Behavior Therapy*, *38*(3), 234–46. doi:10.1016/j.beth.2006.08.003
- Hollon, S. D., Stewart, M. O., & Strunk, D. (2006). Enduring effects for cognitive behavior therapy in the treatment of depression and anxiety. *Annual Review of Psychology*, 57, 285–315. doi:10.1146/annurev.psych.57.102904.190044
- Judd, L., Akiskal, H., & Maser, J. (1998). Major depressive disorder: a prospective study of residual subthreshold depressive symptoms as predictor of rapid relapse. *Journal of Affective Disorders*, 50, 97–108.
- Linehan, M. M., Armstrong, H. E., Suarez, A., Allmon, D., & Heard, H. L. (1991). Cognitivebehavioral treatment of chronically parasuicidal borderline patients. *Archives of General Psychiatry*, 48(12), 1060–1064.
- Mueller, T. I., Leon, A. C., Keller, M. B., Solomon, D. A., Endicott, J., Coryell, W., ...
  Maser, J. D. (1999). Recurrence after recovery from major depressive disorder during 15
  years of observational follow-up. *American Journal of Psychiatry*, 156(7), 1000–1006.
- Paykel, E. S., Ramana, R., Cooper, Z., Hayhurst, H., Kerr, J., & Barocka, A. (1995). Residual symptoms after partial remission: An important outcome in depression. *Psychological Medicine*, 25(6), 1171–1180.
- Piet, J., & Hougaard, E. (2011). The effect of mindfulness-based cognitive therapy for prevention of relapse in recurrent major depressive disorder: A systematic review and meta-analysis. *Clinical Psychology Review*, *31*(6), 1032–40. doi:10.1016/j.cpr.2011.05.002
- Rudd, M. D. (2000). The suicial mode: A cognitive behavioral model of suicidality. *Suicide* and Life-Threatening Behavior, 30, 18 – 33.

- Rudd, M. D. (2006). Fluid vulnerability theory: A cognitive approach to understanding the process of acute and chronic suicide risk. In T. E. Ellis (Ed.), *Cognition and suicide: Theory, research, and therapy* (pp. 355 368). Washington, DC: American Psychological Association.
- Rudd, M. D., Joiner, T. E., & Rajab, M. H. (2001). *Treating suicidal behaviour. An effective time-limited approach*. New York, NY: Guildford Press.
- Rudd, M. D., Schmitz, B., McClenen, R., Joiner, T., Elkins, G., & Claassen, C. (in press). The Suicide Cognitions Scale: A suicide specific measure of hopelessness. *Psychological Assessment*.
- Scher, C. D., Ingram, R. E., & Segal, Z. V. (2005). Cognitive reactivity and vulnerability: empirical evaluation of construct activation and cognitive diatheses in unipolar depression. *Clinical Psychology Review*, 25(4), 487–510. doi:10.1016/j.cpr.2005.01.005
- Segal, Z. V, Bieling, P., Young, T., MacQueen, G., Cooke, R., Martin, L., ... Levitan, R. D. (2010). Antidepressant monotherapy vs sequential pharmacotherapy and mindfulness-based cognitive therapy, or placebo, for relapse prophylaxis in recurrent depression. *Archives of General Psychiatry*, 67(12), 1256–64.

doi:10.1001/archgenpsychiatry.2010.168

- Segal, Z. V, Gemar, M., & Williams, S. (1999). Differential cognitive response to a mood challenge following successful cognitive therapy or pharmacotherapy for unipolar depression. *Journal of Abnormal Psychology*, *108*(1), 3–10.
- Segal, Z. V, Kennedy, S., Gemar, M., Hood, K., Pedersen, R., & Buis, T. (2006). Cognitive Reactivity to Sad Mood Provocation and the Prediction of Depressive Relapse, *63*, 749– 755.
- Segal, Z. V, Williams, J. M. G., & Teasdale, J. D. (2002). *Mindfulness-based cognitive therapy for depression: a new approach to preventing relapse*. New York: Guildford Press.

- Segal, Z. V, Williams, J. M., Teasdale, J. D., & Gemar, M. (1996). A cognitive science perspective on kindling and episode sensitization in recurrent affective disorder, 26, 371–380.
- Solomon, D. A., Keller, M. B., Leon, A. C., Mueller, T. I., Lavori, P. W., Shea, T., ... Endicott, J. (2000). Multiple recurrences of major depressive disorder. *American Journal* of Psychiatry, 157, 229–233.
- Ten Have, M., De Graaf, R., Van Dorsselaer, S., Verdurmen, J., Van't Land, H., Vollebergh,
  W., & Beekman, A. (2009). Incidence and course of suicidal ideation and suicide attempts in the general population. *Canadian Journal of Psychiatry*, 54(12), 824–833.
- Van Orden, K. A., Witte, T. K., Cukrowicz, K. C., Braithwaite, S. R., Selby, E. A., & Joiner, T. E. (2010). The interpersonal theory of suicide. *Psychological Review*, *117*(2), 575–600. doi:10.1037/a0018697
- Williams, J. M. G., Barnhofer, T., Crane, C., & Beck, A. T. (2005). Problem solving deteriorates following mood challenge in formerly depressed patients with a history of suicidal ideation. *Journal of Abnormal Psychology*, *114*(3), 421–31. doi:10.1037/0021-843X.114.3.421
- Williams, J. M. G., Crane, C., Barnhofer, T., Brennan, K., Duggan, D. S., Fennell, M. J. V, ...
  Russell, I. T. (2014). Mindfulness-based cognitive therapy for preventing relapse in
  recurrent depression: A randomized dismantling trial. *Journal of Consulting and Clinical Psychology*, 82(2), 275–86. doi:10.1037/a0035036
- Williams, J. M. G., Crane, C., Barnhofer, T., Van der Does, A. J. W., & Segal, Z. V. (2006).
  Recurrence of suicidal ideation across depressive episodes. *Journal of Affective Disorders*, 91(2-3), 189–94. doi:10.1016/j.jad.2006.01.002
- Williams, J. M. G., Russell, I. T., Crane, C., Russell, D., Whitaker, C. J., Duggan, D. S., ...
  Silverton, S. (2010). Staying well after depression: trial design and protocol. *BMC Psychiatry*, 10, 23. doi:10.1186/1471-244X-10-23

Williams, J. M. G., Van der Does, A. J. W., Barnhofer, T., Crane, C., & Segal, Z. S. (2008).
Cognitive Reactivity, Suicidal Ideation and Future Fluency: Preliminary Investigation of a Differential Activation Theory of Hopelessness/Suicidality. *Cognitive Therapy and Research*, 32(1), 83–104. doi:10.1007/s10608-006-9105-y

Table 1

Sociodemographic Characteristics

Variable	MBCT	CPE	TAU
	(n = 77)	(n = 78)	(n = 39)
Female, <i>n</i> (%)	57 (74)	60 (76)	28 (71)
Age at enrolment, M (SD)	44.5 (11.8)	43.1 (12.2)	43.2 (12.5)
Ethnicity (Caucasian), <i>n</i> (%)	74 (96)	72 (92)	39 (100)
Employed at enrollment, $n(\%)$	52 (67)	45 (57)	26 (66)
Age of first onset of major depression, $M(SD)$	19.3 (8.5)	20.2 (10.6)	21.4 (12.6)
Number of previous episodes, $M(SD)$	8.0 (7.4)	7.0 (3.2)	6.4 (4.7)
Antidepressant use at enrollment, $n$ (%)	31 (40)	33 (42)	14 (35)
SCID Current or past alcohol abuse/substance			
abuse/dependence, $n$ (%)	13 (16)	12 (15)	8 (20)
SCID Current or past anxiety disorder, $n$ (%)	28 (36)	31 (39)	15 (38)

*Note*. SCID = Structured Clinical Interview for DSM-IV Axis I.

# Table 2

	Pre-treatment	Post-treatment
BDI, $M(SD)$		
MBCT ( $n = 52$ )	9.2 (7.4)	9.0 (5.4)
CPE $(n = 54)$	10.6 (10.0)	9.0 (6.5)
TAU ( $n = 26$ )	9.0 (6.8)	11.4 (6.3)
SCS, M(SD)		
MBCT ( $n = 52$ )	36.6 (14.3)	28.1 (7.0) <sup>a</sup>
CPE $(n = 54)$	32.4 (9.1)	29.5 (9.7)
TAU ( $n = 26$ )	32.2 (9.4)	30.4 (6.6)

Means and Standard Deviations of BDI and SCS Scores at Pre- and Post-Treatment

*Note.* <sup>a</sup> Simple main effects with pre-treatment score p < .001, all other simple main effects p > .05.

# **Figure Captions**

*Figure 1*. Scatterplot and regression lines of BDI and SCS scores by group at post-treatment. The top panel shows scores of all participants with past suicidality (MBCT: n = 77, CPE: n = 78, TAU: n = 39). The bottom panel shows scores and regression lines in groups of participants with BDI-II >/= 3 and </= 28 (MBCT: n = 52, CPE: n = 53, TAU: n = 26).

